Author's response to reviews

Title: Tuberculosis among health care workers in KwaZulu-Natal, South Africa: a retrospective cohort analysis

Authors:

Carrie Tudor (ctudor2@jhu.edu)
Martie Van der Walt (martie.vanderwalt@mrc.ac.za)
Bruce Margot (bruce.margot@kznhealth.gov.za)
Susan E Dorman (dsusan1@jhmi.edu)
William K Pan (william.pan@duke.edu)
Gayane Yenokyan (gyenokya@jhsph.edu)
Jason E Farley (jfarley1@jhu.edu)

Version: 2
Date: 22 November 2013

Author's response to reviews: see over
November 22, 2013

Dear BMC Public Health Editors:

Thank you for your prompt attention to our manuscript titled *Tuberculosis among health care workers in KwaZulu-Natal, South Africa: a retrospective cohort analysis* submitted yesterday. We greatly appreciate your feedback.

Per your request, we have copied and pasted the abstract into the webpage on the “manuscript details” tab again. Additionally, we have moved the “key words” from the title page to the bottom of the abstract on page 2 of the manuscript and a revised version of the manuscript has been uploaded.

Please do not hesitate to contact me with any questions.

Best regards,

[Signature]

Carrie Tudor, PhD, MPH, RN